Author's response to reviews

Title: Valproic acid is associated with cognitive decline in HIV-infected individuals: a clinical observational study

Authors:

Lucette A Cysique (lcysique@ucsd.edu)
Paul Maruff (pmaruff@cogstate.com)
Bruce J Brew (b.brew@unsw.edu.au)

Version: 6 Date: 30 November 2006

Author's response to reviews:

Cover letter for the final revised version of MS: 1704500403112957
Valproic acid is associated with cognitive decline in HIV-infected individuals: a clinical observational study
Lucette A Cysique, Paul Maruff and Bruce J Brew

San Diego, November 30th 2006

Dear Dr Jo Appleford
Senior Assistant Editor
BMC-series journals

We are re-submitting the final version of the manuscript including the format changes requested by the editor board.

Thank you for your consideration,
Lucette Cysique, Ph.D.